
%% Emacs mode header: -*-Latex-*- source. %
%% This about the LyX is not true anymore.
%% LyX 2.0.4 created this file. For more info, see http://www.lyx.org/.
%% Do not edit unless you really know what you are doing.
%% History: This project started about 2012/ 2013...
%% 2018: To Do: delete list of tables...

\documentclass[12pt,authoryear]{elsarticle}
% \documentclass[preprint,authoryear,12pt,3p]{elsarticle}
% \usepackage[main=english,czech]{babel}
% \usepackage[latin9]{inputenc}

% [figuresonly,nomarkers,nofiglist]
% \usepackage{url}
% \captionsetup[\ref{tab:parameters}]{list=no}
\captionsetup{list=no}
\renewcommand{\listtablename}{} % \nopagebreak[4]} % {Tables}
\long\def\comment#1{}
\def\citen#1{#1} % works together with LaTeX/Emacs/ispell/ TEXSKIP1
% these funny {bra}{c}{k}{et}s were here to fool the spell-checker :-)
% \newcommand{\hl}[1]{\colorbox{yellow}{#1}}
\newcommand{\hl}[1]{#1}
%\long\def\commentit#1{{\tt #1}}
\def\continue{\par {\bf To be continued...}\par}

\begin{document}
% \listoftables*
\renewcommand*\familydefault{\sfdefault}
\normalfont

%\newcommand{\cyrsf}{}
%\newcommand{\cyrsf}{\fontencoding{OT2}\selectfont\textcyrsf}

% \cortext[Corresp]{Corresponding authors;
% \textit{Cell phone:} +420 777 009 314}
% {+420 728 045 709}

% \address[Academy]{Institute of Experimental Medicine, Academy of
% Sciences of the Czech Republic, V\'\ide\v nsk\'a~1083, 142~20, Praha~4,
% Czech Republic}

% \address[Max]{Max Planck Institute for the Physics of Complex Systems,
% \citen{Noethnitzer}\citen{Strasse} 38, 01187 Dresden, Germany}
% \"o, \ss, etc
% $<$Marsalek@pks.mpg.de$>$.}

% {VoIP02 +420 910 116629.}
% {VoIP1 +420 910 806 204.}
% {Cell1 +420 777 009 314.}
% {Cell2 +49 170 989 7189.}
% {Fax-CZ +420 224 965 916.}

\subsection*{Abbreviations and symbols}
% We use following notation, conforming with \citet{Bures2013} and
% \citet{Sanda2012}. Variables are with small letters, constants with
% capitals and Roman indices and with the exception
% my TeX definitions...
% ---
% \def\cv{C\srom V}$\cv$,~coefficient of variation;
% \def\fsound{f\srom S}$\fsound$,~$f$~sound frequency; 

\comment{ 
%
Even though experimental evidence have changed the theories of how the
MSO computes its output in the decade from 2000 to 2010, for review
see \cite{Grothe2010} and a quantitative view of this is still missing
\cite{Michelet2012}. The neural computation implemented by the MSO
neural circuit is not entirely described to date.
%
} %\EOC

% The neural circuit in the MSO
% Very-raw-paragraph: {\bf In theory}

% It has been reported
% (interim citation: LSO \cite{Tollin2003}), 
% (interim citation: MSO \cite{Grothe2010}).

% After we thoroughly studied mathematical description of our model of
% the MSO circuit, we arrived to observations, which we present here in
% the following section of results.

% The degree of synchronization is measured by the vector strength
% measure, \eqref{eq:M-VS}.

\comment{
Projects of 2013/ 2014: M00176 ``Co-operation in biomedicine and
electronics'' and ``Support of individual careers of prospective
academic workers'' to Z.B. at the College of Polytechnics Jihlava
and by a graduate students' research program of Charles University
% PM.: S{V}V - 2014 - 260033, S{V}V - 2015 - 260157,
Grant support was by: {P}Q{R}S to Z.B. at the College of Polytechnics
Jihlava,
T{U}V at the Czech Academy of Sciences,
and/or WX{Y}Z at the CIIRC,
and by:
Multilateral scientific and
technological cooperation in the Danube region, No. DS-2016-0026,
Plasticity of (spa)tial processing in normal and (c)ochlear (i)mplant
hearing (acronym SPACI) to P.M.}

% and by graduate students' research program of Charles
% University, S{V}V in years 2017-2019 no. 260371
\comment{We thank Denisa X. Svobodova for English language copy-editing.}
% and by internal grant of the Max-Planck Society to P.M.
% http://en.wikipedia.org/wiki/Superior_olivary_complex
\comment{
 % \citen{Kulesza}, R. J., Jr. (2007). "Cyto{-}architecture of the
 % human superior olivary complex: medial and lateral superior olive."
 % Hear Res 225(1-2): 80-90.
} % \EOC 2013, ..., 2019
